Author's response to reviews

Title: Experimental infection of Balb/c nude mice with Hepatitis E virus

Authors:

Fen Huang (huangfen6789@163.com)
Wen Zhang (z0216wen@sjtu.edu.cn)
Ga Gong (gongga@sjtu.edu.cn)
Congli Yuan (ycl@sjtu.edu.cn)
Yijia Yan (y_yan@sjtu.edu.cn)
Shixing Yang (yangshixing@sjtu.edu.cn)
Li Cui (lcui@sjtu.edu.cn)
Jianguo Zhu (Zhu_jg@sjtu.edu.cn)
Zhibiao Yang (zbyang@sjtu.edu.cn)
Xiugo Hua (hxg@sjtu.edu.cn)

Version: 8 Date: 21 May 2009

Author's response to reviews:

Dear Editors,

Thank you for your letter. We had changed the authors’ names in the submission system and changed the format of all pictures. The detail of Reference 30 has been added. We also changed the order of Reference 32 and Reference 33. We removed the line numbers of the manuscript. Thank you very much for your attention again. Best Regards.
Sincerely yours,
Xiuguo Hua